\begin{table}%t1
\caption{\label{tab1}List of target INTEGRAL sources observed with GMRT.}
\small%\centerline
{
\begin{tabular}{lllllll}
\hline
\hline
Source           &Type  &Integral    &Variable                      &X-ray      &X-ray/optical/&No. of sources \\
                 &      &Pos. Unc.   &100~s--1~ks                     &Flux       &UV/IR/Radio   &in the      \\                 &      &1.6$\sigma$ &$\frac{\rm IBIS}{\rm ISGRI}$  &15--40~keV &sources in  &X-ray error circ.       \\
                 &      &            &                              &(mCrab)    &X-ray error circ.   &       \\
\hline
\noalign{\smallskip}
IGR~J00370$+$6122  &HMXB$^{1,2}$   &2$'$        &Yes         &--         &BD$+$6073                     &\\
IGR~J01363$+$6610  &HMXB$^{2}$       &2$'$        &Yes         &17        &HD9603                        &\\
IGR~J16167$-$4957  &--      &6$'$$'$     &--           &2         &--                             &\\
IGR~J16195$-$4945  &HMXB(?)$^{*,10}$   &16$'$$'$    &--           &--         &HD146628                      &\\
IGR~J16207$-$5129  &--      &2$'$        &--           &--         &HD146803                      &\\
IGR~J16358$-$4726  &LMXB(?)$^{3,4}$   &0.6$'$$'$   &Yes         &20--50     &2MASS J163553$-$472539        &\\
                   &Pulsar$^{5}$ &            &            &4.63      &                              &\\
IGR~J16393$-$4643  &HMXB(?)$^{*}$&2$'$        &Yes         &3         &                           &20$^{**}$\\
                   &Pulsar$^{5,6,7,11}$ &            &            &          &                              &\\
IGR~J16558$-$5203  &--      &8$'$$'$     &--           &--         &1RXS J165605$-$520345         &\\                   &       &            &            &          &USNO-B1.0 0379$-$00008129     &\\      
IGR~J17195$-$4100  &--      &8$'$$'$     &Yes         &--         &1RXS J171935$-$410054         &\\
                   &       &            &            &          &USNO-B1.0 0489$-$00511283     &\\
IGR~J17200$-$3116  &--      &9$'$$'$     &Yes         &--         &1RXS J172006$-$311702         &\\
IGR~J17252$-$3616  &HMXB$^{8}$&2$'$        &--           &--         &IRAS 17220$-$3615             &\\                   &Pulsar$^{*}$&            &            &          &NVSS J172510$-$361614         &\\                   &       &            &            &          &HD319824                      &\\
IGR~J17254$-$3257  &--      &14$'$$'$        &--           &--         &1RXS J172525$-$325717         &\\                   &       &            &            &          &USNO-B1.0 0570$-$00727635     &\\
IGR~J17285$-$2922  &XB$^{9}$&2$'$       &Yes         &--         &IRAS 17252$-$2922             &\\
                   &       &            &            &          &[T66b]320                     &\\
IGR~J17303$-$0601  &LMXB$^{*}$&7$'$$'$        &Yes         &--      &H1726$-$058                   &\\                   &       &            &            &          &USNO$-$A2.0 0825$-$10606993   &\\
                   &       &            &            &          &1RXS J173021.5$-$055933       &\\
IGR~J17456$-$2901  &--      &1$'$        &--           &--         &1LC G359.923$-$00.013   &90$^{**}$ \\
IGR~J17460$-$3047  &--      &2$'$        &--           &--         &                        &80 IR$^{+}$ sources$^{**}$\\
IGR~J17464$-$3213  &BHC$^{*}$&0.5$'$    &Yes         &60        &H1743$-$322                   &\\
                   &LMXB$^{3}$&         &            &          &2MASS 17461525-3213542 &\\
                   &          &         &            &          &USNO-A2.0 0525-294112269         &\\
IGR~J17475$-$2822  &Sgr B2$^{5}$&2--3$'$    &--          &--         &                     &200 $^{**}$\\
IGR~J17488$-$3253  &--      &12$'$$'$    &Yes         &--         &1RXS J174854.7$-$325444       &\\
IGR~J18027$-$2016  &Pulsar$^{6}$ &1$'$        &--           &4.06      &HD312525                      &\\
                   &       &            &            &          &1LC G000.683$-$0.035          &\\
                   &       &            &            &          &IRAS 17594$-$2021             &\\
IGR~J18406$-$0539  &--      &2--3$'$    &--           &--          &IRAS 18379$-$0546              &\\                   &       &            &            &          &AX J1840.4$-$0537         &\\
                   &       &            &            &          &NVSS J184037$-$054317         &\\
                   &       &            &            &          &GSC2.2\\
IGR~J18450$-$0435  &--      &2--3$'$    &--           &--          &IRAS 18422$-$0437              &\\
                   &       &            &            &          &PMN J1845$-$0433              \\
IGR~J18490$-$0000  &--      &2--3$'$    &--           &--         &--                              \\
\hline
\end{tabular}}
\smallskip
$^{1}$ High mass X-ray binary, $^{2}$ Reig {et~al.} (\cite{Reig2005}), $^{3}$ Bird {et~al.} (\cite{Bird2004}), $^{4}$ Low mass X-ray binary, $^{5}$
Revnivtsev {et~al.} (2004), $^{6}$ Lutovinov {et~al.} (\cite{Lutovinov2005}), $^{7}$ Boudaghee {et~al.} (2005), $^{8}$ Zurita {et~al.} (\cite{Zurita2005}), 
$^{9}$ X-ray binary, $^{10}$ Sidoli {et~al.} (\cite{Sidoli2005}), $^{11}$ Soldi {et~al.} (\cite{Soldi2005}).\\ 
$^{*}$ {\tt http://isdc.unige.ch/$\sim$rodrigue/html/igrsources.html}, $^{**}$ field sources, $^{+}$ infra red. 
%\vspace*{4mm}
 \end{table}